Author’s response to reviews

Title: Better clinical outcome with direct oral anticoagulants in hospitalized heart failure patients with atrial fibrillation

Authors:

Akiomi Yoshihisa (mgrgh501@ybb.ne.jp)
Yu Sato (yu-sato@fmu.ac.jp)
Shunsuke Watanabe (sw232304@fmu.ac.jp)
Yuki Kanno Yuki Kanno (yuki-k@fmu.ac.jp)
Mai Takiguchi (t-mai@fmu.ac.jp)
Akihiko Sato (akihiko@fmu.ac.jp)
Tetsuro Yokokawa (yokotetu@fmu.ac.jp)
Shunsuke Miura (eggcello@fmu.ac.jp)
Satoshi Abe (abe@fmu.ac.jp)
Takamasa Sato (takamasa@fmu.ac.jp)
Satoshi Suzuki (ssatoshi@fmu.ac.jp)
Masayoshi Oikawa (moikawa@fmu.ac.jp)
Atsushi Kobayashi (koba-a@fmu.ac.jp)
Takayoshi Yamaki (ta-yama@fmu.ac.jp)
Hiroyuki Kunii (hkunii@fmu.ac.jp)
Kazuhiko Nakazato (nakazato@fmu.ac.jp)
Hitoshi Suzuki (shitoshi@fmu.ac.jp)
Shu-ichi Saitoh (sais@fmu.ac.jp)
Yasuchika Takeishi (takeishi@fmu.ac.jp)

Version: 3 Date: 05 Dec 2017
Author’s response to reviews:

Reply to editorials

Thank you for your kind e-mail and helpful comments on our manuscript (BCAR-D-17-00404R2), which we have revised accordingly.

Comment 1:

Unfortunately we are unable to accept the Authorship update form in its current form. Please note that we require the form to be completed by all authors, including those that are being removed, and must be completed using a full signature (not just initials). Please note we also require the 4th column to be edited in order to indicate the changes being made. As such please could you provide a new form with these changes included.

Author’s Response 1:

In line with your advices, we have corrected the authorship update form, and submitted. Additionally, we sent with this e-mail.

Comment 2:

Please ensure that the information regarding Ethics approval and consent to participate is also included within the declarations section.

Author’s Response 2:

In line with your advices, we have corrected this point.
Comment 3:
Please remove the labels 'Figure 1', 'Figure 2', 'Figure 3' from the figure files.

Author's Response 3:
In line with your advices, we have corrected this point.

Comment 4:
Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Author's Response 4:
In line with your advices, we have corrected this point.

Comment 5:
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Author's Response 5:
We have submitted clean manuscript.
Thank you again for your valuable comments on our manuscript.